Author's response to reviews

Title: The DYD-RCT protocol: an on-line randomised controlled trial of an interactive computer-based intervention compared with a standard information website to reduce alcohol consumption among hazardous drinkers.

Authors:

Elizabeth Murray (elizabeth.murray@pcps.ucl.ac.uk)
Jim McCambridge (Jim.McCambridge@lshtm.ac.uk)
Zarnie Khadjesari (z.khadjesari@pcps.ucl.ac.uk)
Ian R White (ian.white@mrc-bsu.cam.ac.uk)
Simon G Thompson (simon.thompson@mrc-bsu.cam.ac.uk)
Christine Godfrey (cg2@york.ac.uk)
Stuart Linke (stuart.linke@candi.nhs.uk)
Paul Wallace (p.wallace@pcps.ucl.ac.uk)

Version: 3 Date: 24 October 2007

Author's response to reviews:

Formatting changes requested by e-mail all completed, namely:

1. The full stop has been removed from the end of the title;
2. The trial registration number has been removed from the title and included in the abstract.
3. The trial registration number is at the end of the abstract.
4. All authors have been placed on one line.
5. The country has been added to the authors' affiliations.
6. the Abstract has been restructured as requested, and changed on the submission page as well.
7. The section numbers have been removed.
8. Manuscript sections are included as requested.
9. A competing interests statement has been included.
10. Reference 11 has been changed as requested.